# The efficacy and safety of triplet regimens based on pomalidomide and dexamethasone for treatment of relapsed/refractory multiple myeloma: a systematic review and meta-analysis

## K.-Y. LIAO, Y. LIU, H. XIONG, X.-M. CHEN, X.-W. ZHANG, C.-L. HUANG

Department of Hematology, Stem Cell Laboratory, Clinical Research Institute, Affiliated Hospital of Southwest Medical University, Luzhou City, Sichuan Province, China

Kunyu Liao and Yang Liu contributed equally to this study

**Abstract.** – OBJECTIVE: Triplet regimens based on pomalidomide and dexamethasone have been applied to treat relapsed/refractory multiple myeloma, but the safety and efficacy are not yet very clear. This meta-analysis aimed at comparing the safety and efficacy of different triplet therapies and analyzing the best therapy regimen.

**MATERIALS AND METHODS:** A comprehensive literature search identified a total of 615 studies, and 22 studies assessing 1,889 subjects met the inclusion criteria of this meta: phase II/III trial, over 2 median lines of prior therapy, and detailed efficacy outcomes like overall response rate (ORR), overall survival, and progression-free survival (PFS). All statistical analyses were performed by Revman version 5.3, and the heterogeneity was tested by  $I^2$  (25% indicating low heterogeneity, 50% moderate, and 75% high). For those with less heterogeneity, fixed-effect model was used. With a significant high heterogeneity, a random-effect model was used.

**RESULTS:** Pooled analysis showed ORR 66.2% across all triplet regimens based on pomalidomide and dexamethasone. Among all triplet regimens, therapy containing bortezomib showed the highest ORR (90.3%), and the one containing elotuzumab showed the lowest ORR (41.2%). The pooled ORRs for the remaining treatment regimens are as follows: cyclophosphamide (70.1%), isatuximab (66.3%), daratumumab (61.2%), clarithromycin (60.0%), pembrolizumab (47.3%). A total of 21 adverse events appeared in the included studies, with neutropenia being the highest incidence of hematologic adverse events (32.1%) and cough being the highest incidence of non-hematologic adverse events (43.3.%).

**CONCLUSIONS:** Three-drug regimens based on pomalidomide and dexamethasone could yield excellent overall response rate to relapsed/ refractory multiple myeloma, but there are still various adverse events; therefore, consequent studies should address these adverse events. Key Words:

Multiple myeloma, Triplet regimen, Overall response, Efficacy, Adverse events.

## Introduction

Multiple myeloma (MM) is a malignant hematological tumor with clonal proliferation of plasma cells in the bone marrow and/or peripheral sites. It is also the second most common hematological malignancy, accounting for 20% of deaths from hematological malignancies<sup>1,2</sup>. MM is characterized by the presence of monoclonal immunoglobulins in the blood or urine, causing anemia, renal insufficiency, extensive bone destruction, hypercalcemia, and recurrent severe infections<sup>3,4</sup>. While research on multiple myeloma has achieved good and encouraging progress over the past few decades, the prognosis of MM remains very poor because of the heterogeneity of clones and the complexity of the genome, and it remains largely considered as an incurable disease. The widespread use of immunomodulators drug (IMiD) and proteasome inhibitors (PI) have prompted the treatment of MM, with complete remission increasing from about 5% to more than 30% and overall survival prolonging from less than 3 years to 5 years<sup>5-7</sup>. Although the introduction of several new drugs has led to improvements, almost all MM patients eventually become relapsed/refractory MM (RRMM). Immune system in RRMM becomes increasingly dysregulated with each treatment, which will lead to aggressive and resistant disease, so the treatment of RRMM remains challenging<sup>8,9</sup>.

It is extremely important to study multidrug combination regimens. Pomalidomide is a third-generation IMiD that mediates the proteasomal degradation of transcription factors by binding to the protein Cereblon in the E3 ubiquitin ligase complex, which exerts powerful, direct antitumor and immune-potentiating effects<sup>10-12</sup>. Although its chemical structure is like that of other IMiDs (thalidomide and lenalidomide), pomalidomide has unique antitumor, antiangiogenic, and immunomodulatory properties<sup>13-15</sup>. Combination therapy with other drugs with different mechanisms improves the prognosis of patients with RRMM, and multi-drug combination regimens provide a feasible approach to overcome the heterogeneity and drug resistance of RRMM and preserve immune system function.

Numerous studies<sup>16,17</sup> have shown a synergistic anti-proliferative activity of pomalidomide with dexamethasone in RRMM patients resistant to lenalidomide. The combination of pomalidomide and dexamethasone has been approved in the United States, Canada, and the European Union for the treatment of RRMM patients. Some clinical trials have used triplet regimens based on pomalidomide and dexamethasone for the treatment of RRMM<sup>18,19</sup>. However, the overall response rate (ORR) of these triplet regimens is not very clear, and most of these studies are clinical trials with small sample sizes, thus the clinical trial results cannot determine the efficacy of the triplet regimens for RRMM. At the same time, the safety and side effects of these triplet regimens also lack summary reports<sup>20,21</sup>. Here, we conducted a meta-analysis of clinical trials to summarize the efficacy and safety of triplet regimens based on pomalidomide and dexamethasone for patients with RRMM.

## **Materials and Methods**

#### Strategy of Literature Search

According to the Preferred Reporting Items for Systematic Reviews and Meta-Analyses Guideline recommended by PRISMA, the literature search in PubMed, MEDLINE, EMBASE, SCOPUS, Cochrane Library and other databases was conducted in May 2021, and the literature language was limited to English. Primary search terms included ("multiple myeloma" OR "MM") AND "pomalidomide" AND "dexamethasone". We included full text of clinical trials without any restrictions concerning age, gender or other factors.

## Inclusion and Exclusion Criteria

The published studies were included in this meta-analysis according to the following criteria: (1) patients with RRMM treated with pomalidomide in combination with dexamethasone; (2) the clinical trials focused on phase II or III; (3) all studies had clear outcome measures, including but not limited to ORR; (4) The study included a description of the incidence of related adverse events. Additionally, we excluded the following literatures: (1) the types of literatures were case report and review article; (2) the studies recruited less than 10 patients; (3) the studies lack of outcome measures. Only most recent studies were included when same publications were based on the same population.

#### Data Extraction

All literatures were carefully screened according to the above criteria. All study data included in this meta-analysis were independently extracted by two experienced reviewers using a standardized data extraction form to avoid selection bias. Disagreements between the two reviewers were resolved by consensus or consultation with a third reviewer. The following information was extracted from each included study: (1) the name of first author; (2) published year of study; (3) the number of patients included in the study; (4) mean age of subjects; (5) therapy schedule based on pomalidomide and dexamethasone; (6) related adverse events; (7) progression-free survival; (8) international staging system (ISS); (9) median lines of prior therapy; (10) overall response rate (95% CI, confidence interval).

#### Statistical Analysis

All data analysis was performed using RevMan 5.3 software (Review Manager Web, The Cochrane collaboration, Copenhagen).  $I^2$ index was applied to calculate the degree of heterogeneity of included studies (25% indicating low heterogeneity, 50% moderate, and 75% high). For those with less heterogeneity, fixed-effect model was used for meta-analysis. If there was significant high heterogeneity, a random-effect model was used to combine the size of effect size in each group. Stratified analysis was performed by dividing the study into treatment groups, forest plots were generated to report the study results, the pooled results are represented by diamonds, and the 95% CI by the lines on both sides of the square. The significance level was set at p < 0.05.

## Results

## Literature Search Results

A total of 615 studies were included from the primary electronic database search: 470 from PubMed, 109 from EMBASE, and 36 from the Cochrane library and SCOPUS. After the assessment for duplicates, 68 studies were excluded. After title and abstract screening, 450 studies were excluded for obvious irrelevance, while 97 articles remained. After full-text screening, 75 studies were excluded because for not examining the disease of interest (n=47), no RRMM definition (n=18), being reviews or meta-analyses (n=7), or not about pomalidomide (n=3). Finally, 22 studies assessing 1,889 subjects were used for this meta-analysis. The study selection procedure is illustrated in Figure 1.

## Main Characteristics of Eligible Studies

Included studies involved a total of 7 treatment regimens, the basic regimens were pomalidomide and dexamethasone (PD), other 7 regimen were cyclophosphamide, bortezomib, clarithromycin, daratumumab, elotuzumab, Isatuximab and pembrolizumab. The most common regimen was PD+ cyclophosphamide (Cyc) with 6 studies<sup>5,6,14,22-24</sup>, followed by 4 trials each of following regimens: PD+ daratumumab (Dara)<sup>15,25-27</sup> and bortezomib (Bort)<sup>20,21,28,29</sup>; 3 trials each with the following regimens: PD+ pembrolizumab (Pem)<sup>30-32</sup>; 2 trials each with the following regimens: PD+ Isatuximab (Isa)<sup>33,34</sup> and Elotuzumab (Elo)<sup>35,36</sup>, only one study<sup>37</sup> applied PD+ clarithromycin (Cla). All patients included in this study had over two prior lines of therapy, and the main previous refractory regimen was Lenalidomide (Table I).

## Response Rate of Triplet Regimens

A total of 22 triplet regimens studies were included, and the pooled analysis of all studies showed an ORR of 66.2% in a random-effect model. The results of subgroup analysis indicated that PD+Bort had the highest ORR of 90.3%, and other ORR are as follows:PD+Cyc (70.1%), PD+I-sa (66.3%), PD+Dara (61.2%), PD+Cla (60.0%), PD+Pem (47.3%) and PD+Elo (41.2%). Detailed data are presented in Figure 2.

#### Adverse Events

There are 21 adverse events which appeared in the included studies, and meta-analysis was per-



Figure 1. PRISMA flow chart of literature search and study selection.

Authors	Year	No. of patients	Median age	Median lines of prior therapy	Refractory to previous regimen	ISS staging (1/2/3)	mSMART(High risk/Standard risk/ Unknown)	Regimen	ORR (%)	PFS (months)
Bringhen et al <sup>33</sup>	2021	154	68	3	Len	64/53/34	24/103/27	PD+Isatuximab	61.5	12.25
Dimopoulos et al <sup>34</sup>	2020	87	66	3	Len	45/32/7	11/63/13	PD+Isatuximab	67.8	12.7
Meletios et al <sup>28</sup>	2020	111	67	2	Len	65/33/13	58/18/35	PD+bortezomib	90	17.8
Paludo et al <sup>29</sup>	2017	50	65.5	2	Len	29/9/12	29/19/2	PD+bortezomib	86	13.7
Richardson et al <sup>20</sup>	2019	281	67	2	Len	149/85/47	137/61/73	PD+bortezomib	82.2	11.2
Sunami et al <sup>21</sup>	2020	12	72	2	Len	10/2	NR	PD+bortezomib	95	16.8
Mark et al <sup>37</sup>	2019	117	63	2	Len	42/37/26	39/28/11	PD+clarithromycin	60	19.2
Baz et al <sup>22</sup>	2016	34	65	≥2	Len	NR	7/8/9	PD+cyclophosphamide	65	12.1
Garderet et al <sup>23</sup>	2018	100	62	NR	NR	67/12/6	69/12/19	PD+cyclophosphamide	82	12.4
Lee et al <sup>24</sup>	2020	55	73.3	≥2	Len	13/30/8	11/33/11	PD+cyclophosphamide	58.2	7.6
Trudel et al <sup>6</sup>	2019	49	66	3	Len 80% Bort35%	29/53/18	9 high	PD+cyclophosphamide	76	6.5
Van Oekelen et al <sup>5</sup>	2020	33	65	3	Len	9/6/3	16/3/15	PD+cyclophosphamide	73	13.3
Soekojo et al <sup>14</sup>	2019	136	67	3	Len, Bort	51/46/35	NR	PD+cyclophosphamide	51.8	10.8
Nooka et al <sup>25</sup>	2019	34	65	3	Len 100%, Bort100%	11/15	26/8/-	PD+daratumumab	58.8	NR
Chari et al <sup>26</sup>	2017	103	64	4	PI+IMiD	NR	65/22/-	PD+daratumumab	60	8.8
Dimopoulos et al <sup>27</sup>	2021	151	67	4	Len 79% Pro 47%	68/50/33	39/64/-	PD+daratumumab	69	12.4
Siegel et al <sup>15</sup>	2020	112	66.5	3	Len	30/53/8	NR	PD+daratumumab	77.7	NR
Hose et al <sup>35</sup>	2020	22	61.5	4	Len	13/5/4	4/15/-	PD+elotuzumab	55	6.4
Dimopoulos et al <sup>36</sup>	2018	60	69	3	Bort100% Len 98%	53/7	NR	PD+elotuzumab	32	10.3
Badros et al <sup>30</sup>	2017	48	64	3	Bort100% Len 100%	NR	30/18/-	PD+pembrolizumab	50	15.6
Mateos et al <sup>31</sup>	2019	125	65	4	Bort 97% Len 95%	45/46/33	28/52/-	PD+pembrolizumab	34	7.8
Matsumoto et al <sup>32</sup>	2021	15	69	2	Bort100% Len 100%	5/8/2	7/8/-	PD+pembrolizumab	47	6.5

Table I. Characteristics of included studies.

PD=Pomalidomide+Dexamethasone; Len=Lenalidomide; Pro=Proeasome; Bort=Bortezomib; ISS=International Staging System; ORR=Overall response rate; PFS=progression-free survival; Msmart=Mayo Stratification of Myeloma and Risk-adapted Therapy; PI=Proteasome inhibitor; IMiD=Immunomodulatory drug; NR=not reported

				Risk Difference	Risk Difference				
Study or Subgroup	Events	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% Cl				
1.1.1 PD+Isatuximab									
Bringhen 2021	95	154	3.0%	0.61 [0.31, 0.92]					
Dimopoulos 2020	59	87	4.3%	0.68 [0.51, 0.85]					
Subtotal (95% CI)			7.4%	0.66 [0.51, 0.81]	•				
Heterogeneity: Tau <sup>2</sup> = 0.00; Chi <sup>2</sup> = 0.13, df = 1 (P = 0.72); l <sup>2</sup> = 0%									
Test for overall effect: Z = 8.7	'5 (P < 0.00	1001)							
4.4.9.00.0									
1.1.2 PD+Bortezomip	100	444	E E 01	0.00.00.00.000	· · · · · · · · · · · · · · · · · · ·				
Meletios 2020	100	111	5.5%	0.90 [0.88, 0.92]					
Paluuo 2017 Dichardeon 2010	43	201	0.470	0.00 [0.03, 0.09]					
Supami 2020	231	12	4.3%	0.82 [0.00, 0.98]	-				
Subtotal (95% CI)		12	20.8%	0.90 [0.85, 0.94]	•				
Heterogeneity: Tau <sup>2</sup> = 0.00; (	$2hi^2 = 24.1$	5. df = 3	(P < 0.00)	01); I² = 88%					
Test for overall effect: Z = 38.01 (P < 0.00001)									
	2	-							
1.1.3 PD+Clarithromycin									
Mark 2019	70	117	3.7%	0.60 [0.37, 0.83]					
Subtotal (95% CI)			3.7%	0.60 [0.37, 0.83]					
Heterogeneity: Not applicabl	e 								
Test for overall effect: $Z = 5.1$	3 (P < 0.00	1001)							
1 1 4 PD+Cyclophosphamid	P								
Baz 2016	22	34	5.3%	0.65 (0.58 .0.72)					
Garderet 2018	82	100	5.5%	0.82 [0.80, 0.84]	•				
Lee 2020	32	55	5.0%	0.58 [0.47, 0.69]					
Soekojo 2019	37	49	5.1%	0.76 [0.66, 0.86]					
Trudel 2019	24	33	5.3%	0.73 [0.67, 0.79]					
Van 2020	70	136	3.3%	0.52 [0.25, 0.78]					
Subtotal (95% CI)			29.4%	0.70 [0.61, 0.79]	•				
Heterogeneity: Tau <sup>2</sup> = 0.01; (	Chi² = 47.88	8, df = 5	(P < 0.00)	001); I² = 90%					
Test for overall effect: $Z = 15$ .	.48 (P < 0.0	10001)							
1.1.5 DD+Daratumumah									
Chari 2017	20	34	5 3%	0.59 (0.52, 0.65)					
Dimonoulos 2021	62	103	4 0%	0.60 [0.32, 0.80]					
Nooka 2019	104	151	31%	0.69 (0.40, 0.09)					
Siegel 2020	87	112	3.8%	0.78 [0.56, 1.00]					
Subtotal (95% CI)			16.2%	0.61 [0.55, 0.67]	•				
Heterogeneity: Tau <sup>2</sup> = 0.00; (	Chi <sup>2</sup> = 2.92,	df = 3 (F	<sup>o</sup> = 0.40);	I <sup>2</sup> = 0%					
Test for overall effect: Z = 19.	.98 (P < 0.0	0001)							
4.4.0 DD. Flat									
1.1.6 PD+Elotuzumap	10	60	4.000	0.00.00.00.0.441					
	19	22	4.9%	0.32 [0.20, 0.44]					
Subtotal (95% CI)	12	22	8.5%	0.41 [0.19, 0.63]	-				
Heterogeneity: Tau <sup>2</sup> = 0.02: (	Chi² = 2.94.	df = 1 (F	P = 0.09):	<sup>2</sup> = 66%					
Test for overall effect: Z = 3.6	6 (P = 0.00	103)	0.007						
1.1.7 PD+Pembrolizumab	1957	3430	15-17-538-5						
Badros 2017	24	48	5.1%	0.50 [0.41, 0.59]					
Mateos 2019	43	125	3.6%	0.34 [0.10, 0.58]					
Matsumoto 2021 Subtotal (05% CI)	₹.	15	5.4%	0.47 [0.44, 0.50]					
Heterononeity: Tau <sup>2</sup> – 0.001	`hi² = 1.47	df = 2/6	P = 0.49	0.47 [0.44, 0.30] P = 0%	Ť				
Test for overall effect: 7 = 33	200 - 1.47, 11 (P < 0.0	un - 2 (r 10001)	- 0.40),	- 0.0					
Total (95% CI)			100.0%	0.66 [0.58, 0.74]	◆				
Heterogeneity: Tau <sup>2</sup> = 0.03; Chi <sup>2</sup> = 958.10, df = 21 (P < 0.00001); l <sup>2</sup> = 98%									
Test for overall effect: Z = 16.61 (P < 0.00001)									
Test for subaroup difference	s: Chi² = 24	49.75. df	= 6 (P < 1	0.00001). I² = 97.6%					

Figure 2. Forest plot of overall response rate for triplet regimens.

formed for all triplet regimens, with the highest incidence of hematologic AEs being neutropenia (32.1%) and the highest incidence of non-hema-

tologic AEs being cough (43.3.%). 19 of 22 studies reported anemia as AEs, with 254 total cases (shown in Table II).

## **Publication Bias**

All included studies reported overall response rate, so ORR was selected for publication bias analysis. Funnel plot results showed that there were no significant publication bias (Figure 3).

#### Therapy Schedule

Most studies set 28 days as a cycle, and the drugs were mainly administered by two routes: intravenous and by oral administration. Drug dose varied slightly in different studies<sup>38-40</sup> and some studies used body area to calculate the dose. Dexamethasone dose was halved to 20 mg in the elderly over 75 years of age in some trials<sup>41,42</sup>. due to some known AEs.

## Discussion

RRMM is currently a major problem for clinicians because of drug resistance, poor treatment prognosis, and serious damage on patients' quality of life<sup>25,29,43</sup>. The insignificant therapeutic effects of previous drugs have prompted the development of the next generation of PI and IMiD, as well as regimen with new mechanisms of action (monoclonal antibodies), which further expands the therapeutic range of RRMM and provides a theoretical basis for the development of combinations of pomalidomide<sup>44-46</sup>. A variety of clinical studies<sup>24,31,32,35,37</sup> have used pomalidomide in combination with different drugs for the treatment of RRMM, and the purpose of using combination regimens is to enhance the therapeutic effect and reduce the occurrence of adverse events using the synergistic effects and non-overlapping toxic effects of different drugs.

At present, the dual combination of pomalidomide and dexamethasone is the most frequently used regimen, and relevant studies<sup>247,48</sup> results show that the treatment effect using the triplet

**Table II.** Adverse events of triplet regimens.

	Related	Total cases	Patients	Heterogeneity test		Pool		
Adverse events	studies			P	р	rate	95%CI	
Hematological adverse events		°			•			
Thrombocytopenia	18	416	1,378	85.2	< 0.001	28.1	22.7-31.5	
Anemia	19	254	1,425	93.1	< 0.001	16.9	14.9-18.6	
Neutropenia	15	378	1,109	90.8	< 0.001	32.1	28.1-36.4	
Leukopenia	17	198	1,372	89.3	< 0.001	15.7	11.9-16.6	
Lymphopenia	18	207	1,195	46.7	0.018	18.9	15.4-19.5	
Non-hematological adverse events								
Fatigue	15	256	1,232	75.3	< 0.001	19.3	18.1-22.5	
Constipation	10	194	1,056	64.2	< 0.001	17.2	16.7-20.1	
Pyrexia	11	392	1,316	43.2	< 0.001	27.3	26.4-30.7	
Pneumonia	13	368	1,344	38.8	< 0.001	26.1	25.1-28.3	
Insomnia	14	347	1,217	79.2	< 0.001	24.9	23.1-29.7	
Diarrhea	9	288	1,008	40.7	< 0.001	26.2	24.6-29.3	
Upper respiratory tract infection	10	274	1,106	31.2	< 0.001	22.3	21.9-25.4	
Dyspnea	7	194	869	11.1	< 0.001	24.6	21.6-25.8	
Nausea	9	306	973	83.4	< 0.001	29.5	28.9-33.1	
Peripheral oedema	10	309	1,116	55.2	< 0.001	26.3	25.1-28.5	
Cough	13	549	1,218	83.1	< 0.001	43.3	41.3-46.3	
Muscle spasms	8	224	968	29.8	< 0.001	22.4	21.0-24.2	
Dizziness	11	352	1,203	18.4	< 0.001	27.6	26.5-30.7	
Headache	8	263	1,180	53.4	< 0.001	21.9	20.1-24.5	
Asthenia	5	118	693	19.9	< 0.001	18.6	16.9-19.8	
Back pain	6	199	849	10.4	< 0.001	24.6	20.9-25.3	



Figure 3. Funnel plot of overall response rate for triplet regimens.

regimens based on pomalidomide and dexamethasone is significantly better than the dual regimens. We performed this meta-analysis to study different triplet regimens, including 22 trials with 1,889 patients. The results indicates that the pooled ORR of triplet regimens is 66.2% and the PFS ranges between 6.4-19.2 months<sup>23,27,34,36</sup>. There is a total of 7 triplet regimens, of which pomalidomide, dexamethasone, and cyclophosphamide were the most used. The results of subgroup analysis show that the combination PD+Bort regimen (pooled ORR 90.3%) appeared to reflect better clinical efficacy, but the number of relevant clinical trials is small<sup>26,28,49</sup>, and whether the study protocol has a definite better therapeutic effect needs to be verified by clinical trials with large samples.

Triplet regimens have been studied in previous studies, Richardson et al<sup>19</sup> conducted a related study on PVD in the treatment of RRMM, and the results showed that the ORR was 82.8%. Further analysis showed that the use of triplet regimens in RRMM patients with cytogenetics could lead to durable and deeper response, significantly improved PFS, and significantly reduced the risk of disease progression and death, indicating that this triple regimen may partially overcome the adverse prognosis of cytogenetic abnormalities, and that pomalidomide can induce response in RRMM patients with lenalidomide resistance<sup>30,50</sup>. The possible mechanism is that pomalidomide has a positive effect on the immune system of patients who fail to respond to lenalidomide by enhancing the activity and number of lenalidomide. Therefore, for patients with previous exposure to lenalidomide or ineffective lenalidomide, no drug class change is required<sup>22,33</sup>.

In term of the therapy schedule, almost all trials<sup>51,52</sup> required multiple courses of medication, scholars<sup>53</sup> applied pomalidomide 4 mg daily as the baseline, but Paludo et al<sup>29</sup> did not endorse this option: they deemed that the dose of 4 mg did not induce better effect than 2 mg. The starting dose of dexamethasone in all studies was 40 mg, scholars<sup>54</sup> halved the dose to 20 mg for patients older than 75 for avoiding adverse events, only few studies<sup>55</sup> did not mention this.

While producing therapeutic effects, triplet therapy also induces AEs, the highest incidence of hematologic AEs is neutropenia (pooled rate 32.1%). There are many types of AEs, and the overall incidence of various AEs is high, indicating that the safety of the triplet regimens needs further study, and subsequent studies can focus on how to effectively manage adverse events.

## Limitations

Limitations of our study are as follows: firstly, the small sample size of the included studies; secondly, conducting crossover trials to compare the optimal triplet regimens based on pomalidomide and dexamethasone for RRMM is imprecise because of heterogeneity between studies in different populations, differences in prior treatment routes, tolerance to prior therapy, ISS stage, cytogenetic risk stratification, and differences in dose and timing. In addition, most analyzed studies<sup>56,57</sup> were not randomized controlled trials, and could have unclear risk of bias, including selection bias and reporting bias; therefore, the overall results of our final meta-analysis are not precise enough. However, in the absence of relevant randomized controlled studies, the results of this study provide a theoretical basis for the relative efficacy of different triplet regimens, and provide useful insights for the subsequent prospective research.

## Conclusions

The results of this meta-analysis showed that three-drug regimens based on pomalidomide and dexamethasone could present excellent over response rate to relapsed/refractory multiple myeloma. However, some adverse events were also recorded in the trials, indicating that a small number of patients had side effects on this treatment regimen. More future studies with good quality are required to further address these adverse events, and to promote the application of triple therapy.

#### **Conflict of Interest**

The authors declare that they have no conflict of interest.

#### Authors' Contributions

KL and YL selected the relevant studies and assessed the data. HX, XC and XZ contributed to the methodological framework. XL and CH revised the manuscript. All authors read and approved the final version of the manuscript.

## Funding

This study was supported by Sichuan department of science and technology (grant No. 20YYJC0940) and Luzhou people's government-Southwest Medical University joint project (grant No. 2017LZXNYD-Z06).

#### References

- Castillo JJ. Evolution of Therapy for Relapsed/ Refractory Multiple Myeloma. J Natl Compr Canc Netw 2019; 17: 1433-1436.
- Cohen YC, Oriol A, Wu KL, Lavi N, Vlummens P, Jackson C, Garvin W, Carson R, Crist W, Fu J, Feng H, Xie H, Schecter J, San-Miguel J, Lonial

S. Daratumumab With Cetrelimab, an Anti-PD-1 Monoclonal Antibody, in Relapsed/Refractory Multiple Myeloma. Clin Lymphoma Myeloma Leuk 2021; 21: 46-54.

- Weisel KC, Scheid C, Zago M, Besemer B, Mai EK, Haenel M, Duerig J, Munder M, Lindemann HW, Seckinger A, Kunz C, Benner A, Hose D, Jauch A, Salwender H, Goldschmidt H. Addition of cyclophosphamide on insufficient response to pomalidomide and dexamethasone: results of the phase II PERSPECTIVE Multiple Myeloma trial. Blood Cancer J 2019; 9: 45.
- Cohen AD. Myeloma: next generation immunotherapy. Hematology Am Soc Hematol Educ Program 2019; 2019: 266-272.
- 5) Van Oekelen O, Parekh S, Cho HJ, Vishnuvardhan N, Madduri D, Richter J, Ip C, Lau K, Florendo E, Mancia IS, Thomas J, Verina D, Chan E, Zarychta K, La L, Strumolo G, Melnekoff DT, Leshchenko VV, Kim-Schulze S, Couto S, Wang M, Pierceall WE, Thakurta A, Laganà A, Jagannath S, Chari A. A phase II study of pomalidomide, daily oral cyclophosphamide, and dexamethasone in relapsed/refractory multiple myeloma. Leuk Lymphoma 2020; 61: 2208-2215.
- 6) Trudel S, Tessoulin B, Jullien M, Blin N, Gastinne T, Mahé B, Dubruille V, Bonnet A, Lok A, Chevallier P, Peterlin P, Garnier A, Guillaume T, Le Bourgeois A, Le Gouill S, Moreau P, Touzeau C. Pomalidomide, cyclophosphamide, and dexamethasone for relapsed/refractory multiple myeloma patients in a real-life setting: a single-center retrospective study. Ann Hematol 2019; 98: 1441-1447.
- Bazarbachi AH, Al Hamed R, Malard F, Harousseau JL, Mohty M. Relapsed refractory multiple myeloma: a comprehensive overview. Leukemia 2019; 33: 2343-2357.
- Landgren O, Sonneveld P, Jakubowiak A, Mohty M, Iskander KS, Mezzi K, Siegel DS. Carfilzomib with immunomodulatory drugs for the treatment of newly diagnosed multiple myeloma. Leukemia 2019; 33: 2127-2143.
- Park SS, Min Byun J, Yoon SS, Kim K, Jung SH, Lee JJ, Min CK. Daratumumab monotherapy for relapsed/refractory multiple myeloma, focussed on clinical trial-unfit patients and subsequent therapy. Br J Haematol 2021; 193: 101-112.
- 10) Ludwig H, Poenisch W, Knop S, Egle A, Schreder M, Lechner D, Hajek R, Gunsilius E, Krenosz KJ, Petzer A, Weisel K, Niederwieser D, Einsele H, Willenbacher W, Melchardt T, Greil R, Zojer N. Ixazomib-Thalidomide-Dexamethasone for induction therapy followed by Ixazomib maintenance treatment in patients with relapsed/refractory multiple myeloma. Br J Cancer 2019; 121: 751-757.
- Lin WY, Wang HH, Chen YW, Lin CF, Fan HC, Lee YY. Gene Modified CAR-T Cellular Therapy for Hematologic Malignancies. Int J Mol Sci 2020; 21: 8655.
- 12) Mateos MV, Sonneveld P, Hungria V, Nooka AK, Estell JA, Barreto W, Corradini P, Min CK, Medve-

dova E, Weisel K, Chiu C, Schecter JM, Amin H, Qin X, Ukropec J, Kobos R, Spencer A. Daratumumab, Bortezomib, and Dexamethasone Versus Bortezomib and Dexamethasone in Patients With Previously Treated Multiple Myeloma: Three-year Follow-up of CASTOR. Clin Lymphoma Myeloma Leuk 2020; 20: 509-518.

- Arcuri LJ, Americo AD. Treatment of relapsed/ refractory multiple myeloma in the bortezomib and lenalidomide era: a systematic review and network meta-analysis. Ann Hematol 2021; 100: 725-734.
- 14) Soekojo CY, Kim K, Huang SY, Chim CS, Takezako N, Asaoku H, Kimura H, Kosugi H, Sakamoto J, Gopalakrishnan SK, Nagarajan C, Wei Y, Moorakonda R, Lee SL, Lee JJ, Yoon SS, Kim JS, Min CK, Lee JH, Durie B, Chng WJ. Pomalidomide and dexamethasone combination with additional cyclophosphamide in relapsed/refractory multiple myeloma (AMN001)-a trial by the Asian Myeloma Network. Blood Cancer J 2019; 9: 83.
- 15) Siegel DS, Schiller GJ, Samaras C, Sebag M, Berdeja J, Ganguly S, Matous J, Song K, Seet CS, Talamo G, Acosta-Rivera M, Bar M, Quick D, Anz B, Fonseca G, Reece D, Pierceall WE, Chung W, Zafar F, Agarwal A, Bahlis NJ. Pomalidomide, dexamethasone, and daratumumab in relapsed refractory multiple myeloma after lenalidomide treatment. Leukemia 2020; 34: 3286-3297.
- 16) Munshi NC, Avet-Loiseau H, Anderson KC, Neri P, Paiva B, Samur M, Dimopoulos M, Kulakova M, Lam A, Hashim M, He J, Heeg B, Ukropec J, Vermeulen J, Cote S, Bahlis N. A large meta-analysis establishes the role of MRD negativity in longterm survival outcomes in patients with multiple myeloma. Blood Adv 2020; 4: 5988-5999.
- 17) Parisi MS, Leotta S, Romano A, Del Fabro V, Martino EA, Calafiore V, Giubbolini R, Markovic U, Leotta V, Di Giorgio MA, Tibullo D, Di Raimondo F, Conticello C. Clinical Benefit of Long-Term Disease Control with Pomalidomide and Dexamethasone in Relapsed/Refractory Multiple Myeloma Patients. J Clin Med 2019; 8: 1695.
- 18) Mele A, Prete E, De Risi C, Citiso S, Greco G, Falcone AP, Sanpaolo G, Mele G, Giannotta A, Vergine C, Reddiconto G, Palazzo G, Sabatelli S, Germano C, Miccolis R, Curci P, Palumbo G, Offidani M, Rizzi R, Cascavilla N, Pastore D, Di Renzo N, Mazza P, Tarantini G, Guarini A, Capalbo S, Specchia G, Greco A, De Francesco R, Sibilla S, Tonialini L, Morciano MR, Pavone V. Carfilzomib, lenalidomide, and dexamethasone in relapsed/ refractory multiple myeloma patients: the real-life experience of Rete Ematologica Pugliese (REP). Ann Hematol 2021; 100: 429-436.
- Richardson PG, Beksaç M, Špička I, Mikhael J. Isatuximab for the treatment of relapsed/refractory multiple myeloma. Expert Opin Biol Ther 2020; 20: 1395-1404.
- 20) Richardson PG, Oriol A, Beksac M, Liberati AM, Galli M, Schjesvold F, Lindsay J, Weisel K, White D, Facon T, San Miguel J, Sunami K, O'Gorman P, Sonneveld P, Robak P, Semochkin S, Schey S,

Yu X, Doerr T, Bensmaine A, Biyukov T, Peluso T, Zaki M, Anderson K, Dimopoulos M; OPTIMISMM trial investigators. Pomalidomide, bortezomib, and dexamethasone for patients with relapsed or refractory multiple myeloma previously treated with lenalidomide (OPTIMISMM): a randomised, open-label, phase 3 trial. Lancet Oncol 2019; 20: 781-794.

- Sunami K, Matsue K, Suzuki K, Takezako N, Shinagawa A, Sakurai S, Tamakoshi H, Biyukov T, Peluso T, Richardson P. Pomalidomide-bortezomib-dexamethasone in relapsed or refractory multiple myeloma: Japanese subset analysis of OPTIMISMM. Cancer Sci 2020; 111: 2116-2122.
- 22) Baz RC, Martin TG 3rd, Lin HY, Zhao X, Shain KH, Cho HJ, Wolf JL, Mahindra A, Chari A, Sullivan DM, Nardelli LA, Lau K, Alsina M, Jagannath S. Randomized multicenter phase 2 study of pomalidomide, cyclophosphamide, and dexamethasone in relapsed refractory myeloma. Blood 2016; 127: 2561-2568.
- 23) Garderet L, Kuhnowski F, Berge B, Roussel M, Escoffre-Barbe M, Lafon I, Facon T, Leleu X, Karlin L, Perrot A, Moreau P, Marit G, Stoppa AM, Royer B, Chaleteix C, Tiab M, Araujo C, Lenain P, Macro M, Voog E, Benboubker L, Allangba O, Jourdan E, Orsini-Piocelle F, Brechignac S, Eveillard JR, Belhadj K, Wetterwald M, Pegourie B, Jaccard A, Eisenmann JC, Glaisner S, Mohty M, Hulin C, Loiseau HA, Mathiot C, Attal M. Pomalidomide, cyclophosphamide, and dexamethasone for relapsed multiple myeloma. Blood 2018; 132: 2555-2563.
- 24) Lee HS, Kim K, Kim SJ, Lee JJ, Kim I, Kim JS, Eom HS, Yoon DH, Suh C, Shin HJ, Mun YC, Kim MK, Lim SN, Choi CW, Kang HJ, Yoon SS, Min CK; Korean Multiple Myeloma Working Party (KMMWP). Pomalidomide, cyclophosphamide, and dexamethasone for elderly patients with relapsed and refractory multiple myeloma: A study of the Korean Multiple Myeloma Working Party (KMMWP-164 study). Am J Hematol 2020; 95: 413-421.
- 25) Nooka AK, Joseph NS, Kaufman JL, Heffner LT, Gupta VA, Gleason C, Boise LH, Lonial S. Clinical efficacy of daratumumab, pomalidomide, and dexamethasone in patients with relapsed or refractory myeloma: Utility of re-treatment with daratumumab among refractory patients. Cancer 2019; 125: 2991-3000.
- 26) Chari A, Suvannasankha A, Fay JW, Arnulf B, Kaufman JL, Ifthikharuddin JJ, Weiss BM, Krishnan A, Lentzsch S, Comenzo R, Wang J, Nottage K, Chiu C, Khokhar NZ, Ahmadi T, Lonial S. Daratumumab plus pomalidomide and dexamethasone in relapsed and/or refractory multiple myeloma. Blood 2017; 130: 974-981.
- 27) Dimopoulos MA, Terpos E, Boccadoro M, Delimpasi S, Beksac M, Katodritou E, Moreau P, Baldini L, Symeonidis A, Bila J, Oriol A, Mateos MV, Einsele H, Orfanidis I, Ahmadi T, Ukropec J, Kampfenkel T, Schecter JM, Qiu Y, Amin H, Vermeulen J, Carson R, Sonneveld P; APOLLO Trial Inves-

tigators. Daratumumab plus pomalidomide and dexamethasone versus pomalidomide and dexamethasone alone in previously treated multiple myeloma (APOLLO): an open-label, randomised, phase 3 trial. Lancet Oncol 2021; 22: 801-812.

- 28) Meletios D, Weisel K, Moreau P, Anderson LD Jr, White D, San-Miguel J, Sonneveld P, Engelhardt M, Jenner M, Corso A, Dürig J, Pavic M, Salomo M, Casal E, Srinivasan S, Yu X, Nguyen TV, Biyukov T, Peluso T, Richardson P. Pomalidomide, bortezomib, and dexamethasone for multiple myeloma previously treated with lenalidomide (OP-TIMISMM): outcomes by prior treatment at first relapse. Leukemia 2020; 35: 1722-1731.
- 29) Paludo J, Mikhael JR, LaPlant BR, Halvorson AE, Kumar S, Gertz MA, Hayman SR, Buadi FK, Dispenzieri A, Lust JA, Kapoor P, Leung N, Russell SJ, Dingli D, Go RS, Lin Y, Gonsalves WI, Fonseca R, Bergsagel PL, Roy V, Sher T, Chanan-Khan AA, Ailawadhi S, Stewart AK, Reeder CB, Richardson PG, Rajkumar SV, Lacy MQ. Pomalidomide, bortezomib, and dexamethasone for patients with relapsed lenalidomide-refractory multiple myeloma. Blood 2017; 130: 1198-1204.
- 30) Badros A, Hyjek E, Ma N, Lesokhin A, Dogan A, Rapoport AP, Kocoglu M, Lederer E, Philip S, Milliron T, Dell C, Goloubeva O, Singh Z. Pembrolizumab, pomalidomide, and low-dose dexamethasone for relapsed/refractory multiple myeloma. Blood 2017; 130: 1189-1197.
- 31) Mateos MV, Blacklock H, Schjesvold F, Oriol A, Simpson D, George A, Goldschmidt H, Larocca A, Chanan-Khan A, Sherbenou D, Avivi I, Benyamini N, lida S, Matsumoto M, Suzuki K, Ribrag V, Usmani SZ, Jagannath S, Ocio EM, Rodriguez-Otero P, San Miguel J, Kher U, Farooqui M, Liao J, Marinello P, Lonial S; KEYNOTE-183 Investigators. Pembrolizumab plus pomalidomide and dexamethasone for patients with relapsed or refractory multiple myeloma (KEYNOTE-183): a randomised, open-label, phase 3 trial. Lancet Haematol 2019; 6: e459-e469.
- 32) Matsumoto M, Suzuki K, Kuroda J, Taniwaki M, Sunami K, Kosugi H, Ando K, Maruyama D, Tobinai K, Kher U, Farooqui M, Liao J, Marinello P, Matsuda K, Koh Y, Shimamoto T, lida S. Pembrolizumab plus pomalidomide and dexamethasone for relapsed or refractory multiple myeloma (KEYNOTE-183): subgroup analysis in Japanese patients. Int J Hematol 2021; 113: 777-784.
- 33) Bringhen S, Pour L, Vorobyev V, Vural F, Warzocha K, Benboubker L, Koh Y, Maisnar V, Karlin L, Pavic M, Campana F, Le Guennec S, Menas F, van de Velde H, Richardson PG. Isatuximab plus pomalidomide and dexamethasone in patients with relapsed/refractory multiple myeloma according to prior lines of treatment and refractory status: ICARIA-MM subgroup analysis. Leuk Res 2021; 104: 106576.
- 34) Dimopoulos MA, Leleu X, Moreau P, Richardson PG, Liberati AM, Harrison SJ, Miles Prince H, Ocio EM, Assadourian S, Campana F, Malinge L, Sémiond D, van de Velde H, Yong K. Isatuximab

plus pomalidomide and dexamethasone in relapsed/refractory multiple myeloma patients with renal impairment: ICARIA-MM subgroup analysis. Leukemia 2020; 35: 562-572.

- 35) Hose D, Schreder M, Hefner J, Bittrich M, Danhof S, Strifler S, Krauth MT, Schoder R, Gisslinger B, Einsele H, Gisslinger H, Knop S. Elotuzumab, pomalidomide, and dexamethasone is a very well tolerated regimen associated with durable remission even in very advanced myeloma: a retrospective study from two academic centers. J Cancer Res Clin Oncol 2020; 147: 205-212.
- 36) Dimopoulos MA, Dytfeld D, Grosicki S, Moreau P, Takezako N, Hori M, Leleu X, LeBlanc R, Suzuki K, Raab MS, Richardson PG, Popa McKiver M, Jou YM, Shelat SG, Robbins M, Rafferty B, San-Miguel J. Elotuzumab plus Pomalidomide and Dexamethasone for Multiple Myeloma. N Engl J Med 2018; 379: 1811-1822.
- 37) Mark TM, Forsberg PA, Rossi AC, Pearse RN, Pekle KA, Perry A, Boyer A, Tegnestam L, Jayabalan D, Coleman M, Niesvizky R. Phase 2 study of clarithromycin, pomalidomide, and dexamethasone in relapsed or refractory multiple myeloma. Blood Adv 2019; 3: 603-611.
- 38) Moreau P, Dimopoulos MA, Yong K, Mikhael J, Risse ML, Asset G, Martin T. Isatuximab plus carfilzomib/dexamethasone versus carfilzomib/dexamethasone in patients with relapsed/refractory multiple myeloma: IKEMA Phase III study design. Future Oncol 2020; 16: 4347-4358.
- 39) Premkumar V, Bhutani D, Lentzsch S. Modern Treatments and Future Directions for Relapsed/ Refractory Multiple Myeloma Patients. Clin Lymphoma Myeloma Leuk 2020; 20: 736-743.
- 40) Alsouqi A, Khan M, Dhakal B, Du L, Harrell S, Hari P, Cornell RF. KD-PACE Salvage Therapy for Aggressive Relapsed Refractory Multiple Myeloma, Plasma Cell Leukemia and Extramedullary Myeloma. Clin Lymphoma Myeloma Leuk 2021; 21: 526-535.
- 41) Bahlis NJ, Dimopoulos MA, White DJ, Benboubker L, Cook G, Leiba M, Ho PJ, Kim K, Takezako N, Moreau P, Kaufman JL, Krevvata M, Chiu C, Qin X, Okonkwo L, Trivedi S, Ukropec J, Qi M, San-Miguel J. Daratumumab plus lenalidomide and dexamethasone in relapsed/refractory multiple myeloma: extended follow-up of POL-LUX, a randomized, open-label, phase 3 study. Leukemia 2020; 34: 1875-1884.
- 42) Schjesvold F, Bringhen S, G Richardson P, Perrot A, Leleu X, Moreau P, A Dimopoulos M, Hulin C, Tekle C, Foster MC, Poole EM, van de Velde H, Facon T. Isatuximab plus pomalidomide and dexamethasone in frail patients with relapsed/refractory multiple myeloma: ICARIA-MM subgroup analysis. Am J Hematol 2021; 96: E423-E427.
- 43) Richardson PG, Attal M, Campana F, Le-Guennec S, Hui AM, Risse ML, Corzo K, Anderson KC. Isatuximab plus pomalidomide/dexamethasone versus pomalidomide/dexamethasone in relapsed/refractory multiple myeloma: ICARIA

Phase III study design. Future Oncol 2018; 14: 1035-1047.

- 44) Shah J, Usmani S, Stadtmauer EA, Rifkin RM, Berenson JR, Berdeja JG, Lyons RM, Klippel Z, Chang YL, Niesvizky R. Oprozomib, pomalidomide, and Dexamethasone in Patients With Relapsed and/or Refractory Multiple Myeloma. Clin Lymphoma Myeloma Leuk 2019; 19: 570-578.
- 45) Wudhikarn K, Mailankody S, Smith EL. Future of CAR T cells in multiple myeloma. Hematology Am Soc Hematol Educ Program 2020; 2020: 272-279.
- 46) Jo JC, Lee HS, Kim K, Lee JJ, Yoon SS, Bang SM, Kim JS, Eom HS, Yoon DH, Lee Y, Shin HJ, Park Y, Lee WS, Do YR, Mun YC, Lee MH, Kim HJ, Kim SH, Kim MK, Lim SN, Cho SH, Park SK, Yi JH, Lee JH, Kim J, Min CK; Korean Multiple Myeloma Working Party (KMMWP). The effectiveness and safety of lenalidomide and dexamethasone in patients with relapsed/refractory multiple myeloma in real-world clinical practice: a study of the Korean Multiple Myeloma Working Party (KM-MWP-151 study). Ann Hematol 2020; 99: 309-319.
- 47) Rocchi S, Tacchetti P, Pantani L, Mancuso K, Rizzello I, di Giovanni Bezzi C, Scalese M, Dozza L, Marzocchi G, Martello M, Barilà G, Antonioli E, Staderini M, Buda G, Petrini M, Cea M, Quaresima M, Furlan A, Bonalumi A, Cavo M, Zamagni E. A real-world efficacy and safety analysis of combined carfilzomib, lenalidomide, and dexamethasone (KRd) in relapsed/refractory multiple myeloma. Hematol Oncol 2021; 39: 41-50.
- 48) Schmidts A, Ormhøj M, Choi BD, Taylor AO, Bouffard AA, Scarfò I, Larson RC, Frigault MJ, Gallagher K, Castano AP, Riley LS, Cabral ML, Boroughs AC, Velasco Cárdenas RM, Schamel W, Zhou J, Mackay S, Tai YT, Anderson KC, Maus MV. Rational design of a trimeric APRIL-based CAR-binding domain enables efficient targeting of multiple myeloma. Blood Adv 2019; 3: 3248-3260.
- 49) Cohen A, Spektor TM, Stampleman L, Bessudo A, Rosen PJ, Klein LM, Woliver T, Flam M, Eshaghian S, Nassir Y, Maluso T, Swift RA, Vescio R, Berenson JR. Safety and efficacy of pomalidomide, dexamethasone and pegylated liposomal doxorubicin for patients with relapsed or refractory multiple myeloma. Br J Haematol 2018; 180: 60-70.

- 50) Attal M, Richardson PG, Rajkumar SV, San-Miguel J, Beksac M, Spicka I, Leleu X, Schjesvold F, Moreau P, Dimopoulos MA, Huang JS, Minarik J, Cavo M, Prince HM, Macé S, Corzo KP, Campana F, Le-Guennec S, Dubin F, Anderson KC; ICARIA-MM study group. Isatuximab plus pomalidomide and low-dose dexamethasone versus pomalidomide and low-dose dexamethasone in patients with relapsed and refractory multiple myeloma (ICARIA-MM): a randomised, multicentre, open-label, phase 3 study. Lancet 2019; 394: 2096-2107.
- 51) Hanna KS, Larson S, Nguyen J, Tu S, Boudreau J, Rose S. Updates in the management of relapsed/ refractory multiple myeloma. J Oncol Pharm Pract 2021; 27: 1477-1490.
- 52) Lei M, Kim EB, Branagan A, Lou U, Zemel M, Raje N. Current management and emerging treatment strategies for multiple myeloma. Rinsho Ketsueki 2019; 60: 1243-1256.
- 53) Piggin A, Prince HM. An evaluation of isatuximab, pomalidomide and dexamethasone for adult patients with relapsed and refractory multiple myeloma. Expert Rev Hematol 2021; 14: 419-427.
- Robak P, Drozdz I, Szemraj J, Robak T. Drug resistance in multiple myeloma. Cancer Treat Rev 2018; 70: 199-208.
- 55) Pierceall WE, Amatangelo MD, Bahlis NJ, Siegel DS, Rahman A, Van Oekelen O, Neri P, Young M, Chung W, Serbina N, Parekh S, Agarwal A, Thakurta A. Immunomodulation in Pomalidomide, Dexamethasone, and Daratumumab-Treated Patients with Relapsed/Refractory Multiple Myeloma. Clin Cancer Res 2020; 26: 5895-5902.
- 56) Suzuki K, Min CK, Kim K, Lee JJ, Shibayama H, Ko PS, Huang SY, Li SS, Ding B, Khurana M, lida S. Carfilzomib, dexamethasone, and daratumumab in Asian patients with relapsed or refractory multiple myeloma: post hoc subgroup analysis of the phase 3 CANDOR trial. Int J Hematol 2021; 114: 653-663.
- 57) Schjesvold F, Robak P, Pour L, Aschan J, Sonneveld P. OCEAN: a randomized Phase III study of melflufen + dexamethasone to treat relapsed refractory multiple myeloma. Future Oncol 2020; 16: 631-641.